Author's response to reviews

Title: A child presenting acute renal failure secondary to a high dose of indomethacin: a case report

Authors:

Felipe Gonzalez (pielvi@ya.com)
Jesus Lopez-Herce (pielvi@ya.com)
Cinta Moraleda (pielvi@ya.com)

Version: 2 Date: 8 July 2008

Author's response to reviews: see over
Dear Sir

We enclose you the article "A child presenting acute renal failure secondary to a high dose of indomethacin: a case report", that has been reformed according to your suggestions

1. I have restructured the title
2. We have included information on the patient’s country, ethnic origin in the case presentation section.
3. I have included a list of abbreviations used in the manuscript and their meanings
4. I have included an acknowledgement section
5. I have not included a figure title and legend section after the reference list because there are not figures in our article
6. The text have been revised by a professional English speaker medical translator

We hope it can be published as Case Report in the Journal of Medical Case Report Journal

This is in our knowledge the first case of acute renal failure in children secondary to the administration of a high dose of indomethacin. Our article also revise and discuss the errors in the systems of prescription and administration of medication.

The final manuscript has been seen and approved by all authors. The article has not been published previously or submitted for publication elsewhere.

Sincerely Yours

Prof. Dr Jesús López-Herce
Correspondence:

Dr Jesús López-Herce
Sección de Cuidados Intensivos Pediátricos. Hospital Gregorio Marañón.
Dr Castelo 47. 28009 Madrid. Spain.
Fax: -34-91-5868018
Tf: -34-91-5290308
E-mail: pielvi@ya.com